Author's response to reviews

Title: The glutamate transport inhibitor DL-Threo-beta-Benzoyloxyaspartic acid (DL-TBOA) differentially affects SN38- and oxaliplatin-induced death of drug-resistant colorectal cancer cells

Authors:

- Elena Pedraz-Cuesta (elenapedraz@gmail.com)
- Sandra Christensen (sandra.christensen@outlook.com)
- Anders A Jensen (aaj@sund.ku.dk)
- Niels F Jensen (nielsfrankjensen@gmail.com)
- Lennart Bunch (Lebu@sund.ku.dk)
- Maria U Romer (maria.unni.romer@gmail.com)
- Nils Brunner (nbr@sund.ku.dk)
- Jan Stenvang (stenvang@sund.ku.dk)
- Stine F Pedersen (stpedersen@bio.ku.dk)

Version: 6
Date: 27 April 2015

Author's response to reviews: see over
Response to Editor’s decision and comments – MS 1601353759153108

24 APRIL 2015

Dear Dr. Rowles,

We are very pleased that our manuscript was accepted for publication, and have made the requested addition, as noted below. A revised version of the manuscript has been uploaded.

Thank you for finding our work suitable for publication in BMC Cancer.

Yours sincerely, on behalf of the authors,

Stine Falsig Pedersen
Professor

Point-by-point response to the Editorial comments:

- Please include details in your Methods section on the source from which the cell lines used in your study were purchased.

This has now been done.